Author's response to reviews

Title: Rupture of Heterologous Pericardial Patch after Aortic Annulus Enlargement

Authors:

   Akimasa Morisaki (m3_514@yahoo.co.jp)
   Yasuyuki Kato (kato20001224@gmail.com)
   Manabu Motoki (m6984787@msic.med.osaka-cu.ac.jp)
   Yosuke Takahashi (ysk@msic.med.osaka-cu.ac.jp)
   Shinsuke Nishimura (s_nishimura07@yahoo.co.jp)
   Toshihiko Shibata (shibata-cvs@zeus.eonet.ne.jp)

Version: 2
Date: 29 January 2014

Author's response to reviews:

1. Please reformat the manuscript title. The title should include the study design, for example "A versus B in the treatment of C: a randomized controlled trial X is a risk factor for Y: a case control study"

   Response: Thank you for comment. I revised the title name including the study design.

2. Please include all the co-authors' email addresses in the Title page.

   Response: Thank you for your comment. I had added all the co-author's email addresses.

3. Please include the Ethnicity of the patient in the Abstract and Case presentation sections.

   Response: I appreciated your comment. I had added the ethnicity of the patient as Japanese.